Title: Obesity is a Negative Predictor of Success after Surgery for Complex Anal Fistula

Authors:

Oliver Schwandner (oliver.schwandner@barmherzige-regensburg.de)

Version: 5 Date: 16 May 2011

Author's response to reviews:

As requested minor changes have been made:

"Acknowledgement" has been changed to "Competing Interests".

Tables - vertical lines have been removed.

Typography: Some formal minor changes have been made.

Best regards
Oliver Schwandner
May 16, 2011